

Fine Pathogen Discrimination within the APL1 Gene Family Protects Anopheles gambiae against Human and Rodent Malaria Species

Christian Mitri, Jean-Claude Jacques, Isabelle Thiéry, Michelle M. Riehle, Jiannong Xu, Emmanuel Bischoff, Isabelle Morlais, Sandrine E. Nsango, Kenneth D Vernick, Catherine Bourgouin

▶ To cite this version:

Christian Mitri, Jean-Claude Jacques, Isabelle Thiéry, Michelle M. Riehle, Jiannong Xu, et al.. Fine Pathogen Discrimination within the APL1 Gene Family Protects Anopheles gambiae against Human and Rodent Malaria Species. PLoS Pathogens, 2009, 5 (9), pp.e1000576. 10.1371/journal.ppat.1000576 . ird-02896969

HAL Id: ird-02896969 https://ird.hal.science/ird-02896969

Submitted on 1 Jun 2021

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

Fine Pathogen Discrimination within the APL1 Gene Family Protects Anopheles gambiae against Human and Rodent Malaria Species

Christian Mitri^{1,2}, Jean-Claude Jacques², Isabelle Thiery², Michelle M. Riehle³, Jiannong Xu^{3¤}, Emmanuel Bischoff¹, Isabelle Morlais⁴, Sandrine E. Nsango⁴, Kenneth D. Vernick^{1,3*}, Catherine Bourgouin^{1,2}

1 Unit of Insect Vector Genetics and Genomics, Department of Parasitology and Mycology, CNRS Unit URA3012: Hosts, Vectors and Infectious Agents, Institut Pasteur, Paris, France, 2 Center for the Production and Infection of Anopheles, Institut Pasteur, Paris, France, 3 Department of Microbiology, University of Minnesota, Saint Paul, Minnesota, United States of America, 4 Laboratoire de Recherche sur le Paludisme, Institut de Recherche pour le Développement IRD-OCEAC, Yaoundé, Cameroun

Abstract

Genetically controlled resistance of Anopheles gambiae mosquitoes to Plasmodium falciparum is a common trait in the natural population, and a cluster of natural resistance loci were mapped to the Plasmodium-Resistance Island (PRI) of the A. gambiae genome. The APL1 family of leucine-rich repeat (LRR) proteins was highlighted by candidate gene studies in the PRI, and is comprised of paralogs APL1A, APL1B and APL1C that share \geq 50% amino acid identity. Here, we present a functional analysis of the joint response of APL1 family members during mosquito infection with human and rodent Plasmodium species. Only paralog APL1A protected A. gambiae against infection with the human malaria parasite P. falciparum from both the field population and in vitro culture. In contrast, only paralog APL1C protected against the rodent malaria parasites P. berghei and P. yoelii. We show that anti-P. falciparum protection is mediated by the Imd/Rel2 pathway, while protection against P. berghei infection was shown to require Toll/Rel1 signaling. Further, only the short Rel2-S isoform and not the long Rel2-F isoform of Rel2 confers protection against P. falciparum. Protection correlates with the transcriptional regulation of APL1A by Rel2-S but not Rel2-F, suggesting that the Rel2-S anti-parasite phenotype results at least in part from its transcriptional control over APL1A. These results indicate that distinct members of the APL1 gene family display a mutually exclusive protective effect against different classes of Plasmodium parasites. It appears that a gene-forpathogen-class system orients the appropriate host defenses against distinct categories of similar pathogens. It is known that insect innate immune pathways can distinguish between grossly different microbes such as Gram-positive bacteria, Gram-negative bacteria, or fungi, but the function of the APL1 paralogs reveals that mosquito innate immunity possesses a more fine-grained capacity to distinguish between classes of closely related eukaryotic pathogens than has been previously recognized.

Citation: Mitri C, Jacques J-C, Thiery I, Riehle MM, Xu J, et al. (2009) Fine Pathogen Discrimination within the APL1 Gene Family Protects Anopheles gambiae against Human and Rodent Malaria Species. PLoS Pathog 5(9): e1000576. doi:10.1371/journal.ppat.1000576

Editor: David S. Schneider, Stanford University, United States of America

Received March 9, 2009; Accepted August 12, 2009; Published September 11, 2009

Copyright: © 2009 Mitri et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This work was supported by NIAID/NIH AI042361, AI073685. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: kenneth.vernick@pasteur.fr

¤ Current address: Department of Biology, New Mexico State University, Las Cruces, New Mexico, United States of America

Introduction

Malaria remains a major global health problems, with more than 300 million estimated cases and more than one million deaths annually [1]. *Plasmodium falciparum* is the most common human malaria parasite in Africa and is responsible for the majority of mortality due to malaria. The vector mosquito, *Anopheles gambiae*, is widespread in sub-Saharan Africa, where it constitutes the major vector of *P. falciparum*. Vector control has been one of the foundations of malaria control. However, recent problems such as the rapid spread of drug-resistant parasites and insecticideresistant mosquitoes have encouraged the development of a new generation of vector-based malaria control tools. Such new vectorbased strategies might be based on the introduction into the vector population of genetic information encoding novel desirable traits that reduce transmission [2], or on exploiting natural selection to promote the expansion of pre-existing desirable traits [3,4].

The high level of *P. falciparum* transmission in Africa, which is sustained in large part by *A. gambiae*, led to the idea that this mosquito species is generally permissive for the development of *P. falciparum*. However, analysis of the field population indicates that genetically-controlled resistance of wild *A. gambiae* mosquitoes to *P. falciparum* in West Africa is relatively frequent [5]. Genetic mapping led to the identification of major-effect loci for *P. falciparum* resistance that cluster within the PRI of the *A. gambiae* genome [4]. This same genomic region was also identified in East African *A. gambiae* as a major locus controlling *P. falciparum* development [6,7]. The identification of common genetic control over susceptibility to *P. falciparum* infection in geographically distant *A. gambiae* populations suggests that a shared and

Author Summary

The African malaria vector mosquito Anopheles gambiae possesses immune mechanisms that can protect it against infection with malaria parasites, which kill more than one million people per year. Much work studying mosquito response to malaria has used model rodent malaria parasites that do not infect people, but are in the same genus and overall share most major features with the human parasites. Here, we show that the immune response used by A. gambiae to protect itself against infection by human and rodent malaria parasites utilizes different immune signaling pathways. A family of proteins called APL1 appears to be responsible for the ability of the mosquito to distinguish between the human or rodent malaria parasites. An individual APL1 relative is required for protection against the human malaria parasite but has no effect against the rodent parasite, and another APL1 relative is required for protection against rodent but not human malaria. This represents the finest ability yet demonstrated of mosquito immunity to distinguish between relatively similar pathogens, and highlights the distinct nature of mosquito response against human as compared to rodent malaria parasites.

widespread resistance mechanism operates in *A. gambiae* throughout sub-Saharan Africa. Such a widespread trait might be an evolutionarily old host-defense mechanism that could have been shaped and maintained by selective pressure from malaria or other similar pathogens.

A candidate gene study of the ~1000 predicted protein coding sequences in the ~15 Mb PRI locus yielded two genes that survived all filter criteria [4]. These two genes, named *APL1* and *APL2*, encode LRR proteins, which is a widespread motif in innate immune proteins among animals and plants [8]. RNAi-mediated gene silencing in an *A. gambiae* laboratory strain indicated that candidate gene *APL1*, but not *APL2*, was highly protective against the development of the rodent malaria parasite, *P. berghei* [4]. Subsequent reannotation of the *APL1* locus revealed the original *APL1* "gene" to be three paralogous members of a gene family, which were then named *APL1A*, *APL1B* and *APL1C* [9]. Genespecific knockdown assays showed that of the three paralogs, only *APL1C* was responsible for the previously observed control over *P. berghei* parasite intensity and infection prevalence.

Here, we functionally analyzed the joint response of APL1 family members during mosquito infection with both human and rodent Plasmodium species. First, using a field assay system we discovered that the action of the APL1 family conferred mosquito resistance against wild P. falciparum. We then transferred the same mosquito assay strain to the laboratory where we challenged mosquitoes with in vitro cultured P. falciparum gametocytes, and observed equivalent APL1 protection as against the wild parasites. This provided a field-to-lab validation trail for the biological fidelity of the assay system, and justified use of the system for detailed laboratory dissection of APL1-based protection. To date, a small number of genes ascertained by laboratory studies have been shown to protect A. gambiae against P. falciparum [10-12]. A unique aspect of the current work is that we instead ascertained candidate P. falciparum resistance genes by initially genetically screening the natural population, and then extracted the trait from field to lab for functional dissection.

We report that of the three *APL1* family members, only paralog *APL1A* displayed a host-protective effect against *P. falciparum*, which we show is mediated by a specific sub-circuit of the Rel2

signalling pathway. In contrast, we extend the observation that only *APL1C* was responsible for control of *P. berghei* infection, which is mediated by the Rell pathway, with another rodent malaria species, *P. yoelii*. Thus, distinct members of the APL1 family of LRR proteins display a mutually exclusive protective effect against different classes of *Plasmodium* parasites, in the context of distinct immune signalling pathways.

Results

APL1 controls the development of natural P. falciparum isolates in A. gambiae

We tested the role of the *APL1* gene family in mosquito susceptibility to *P. falciparum* using wild parasites and a colonized strain of *A. gambiae*. A common *APL1* family domain is highly enough conserved at the nucleic acid level that a double-stranded RNA, *dswAPL1*, efficiently silences all three paralogs. We used *dswAPL1* to analyze the summed role of the *APL1* family on the development of wild *P. falciparum* in the recently colonized Ngousso strain of *A. gambiae*. Mosquitoes injected with *dswAPL1* were challenged by feeding on the blood of naturally-infected *P. falciparum* gametocyte carriers in Cameroon. The effect of *APL1* silencing on *P. falciparum* development was measured by determining oocyst number 7–8 d post-bloodmeal. Silencing of the entire *APL1* family resulted in a significant increase of infection prevalence (Figure 1; p = 0.006).

All gene silencing experiments presented herein were done as at least three biological replicates of at least 30 dissected mosquitoes per replicate (see Methods). Replicates were first analyzed independently to determine if there were significant differences between replicates within treatments: if not, data were pooled and analyzed statistically; if so, pooling is not statistically valid and replicates were instead analyzed independently and resulting p values were combined by meta-analysis (see Methods). The threshold of significance was defined conservatively as p = 0.01. Statistical details are summarized in Supplementary Tables S2 and S3. Infection prevalence is the fraction of mosquitoes carrying at least one oocyst, while parasite intensity is the number of oocysts per mosquito determined only in the subset of mosquitoes with ≥ 1 oocyst (see Methods). The APL1 knockdown phenotype controlled only infection prevalence, with no detectable effect on oocyst intensity (p-value for intensity = 0.926). This result indicates that wild-type expression of the APL1 gene family has a host-protective effect against natural genotypes of P. falciparum circulating in the field population.

Paralog *APL1A* controls a developmental blockade of *P. falciparum*

Having found by knockdown of the entire gene family that *APL1* is protective against *P. falciparum*, we wished to determine the relative phenotypic contribution of the *APL1* paralogs. For this, we moved the experimental system to the laboratory to take advantage of the ease and reproducibility of *P. falciparum in vitro* gametocyte culture for infection of the same Ngousso mosquito strain used in Cameroon.

There is sufficient nucleotide sequence divergence outside the *APL1* common domain to permit paralog-specific gene silencing [9]. Targeting non-conserved regions, we produced specific dsRNAs that diminish the expression of one paralog without affecting the expression of the other two (Figure 2A). Of the three genes, only the *APL1A* knockdown displayed a phenotype after *P. falciparum* challenge, with significantly higher prevalence of *P. falciparum* infection as compared to controls (Figure 2B; p = 0.001). Similar to the result described above for *dswAPL1* and wild *P.*



Figure 1. APL1 protects A. gambiae against wild P. falciparum. A) Expression of the APL1 gene family detected by RT-PCR using primers that amplify a conserved region shared by all three APL1 genes. Mosquitoes were injected with a dsRNA directed against the shared region to simultaneously silence the three APL1 paralogs (dswAPL1), or with an irrelevant control dsRNA (dsLacz). In this and subsequent figures, the injected dsRNA is indicated above the gel columns and the transcript detected by RT-PCR is indicated to the right of the relevant gel row. Decreased signal in row wAPL1 indicates that dswAPL1 efficiently silences the entire APL1 family as compared to dsLacZinjected controls. Detection of mRNA for ribosomal protein S7 (rpS7) was used as an internal reference. B) Silencing of the APL1 gene family permits significantly higher infection prevalence in mosquitoes fed on wild P. falciparum gametocytes from the natural population in Cameroon. There was no detectable effect of dswAPL1 on parasite intensity. Infection prevalence effects are depicted by pie charts rather than histograms of parasite number because the significant difference is between infected or uninfected states, not parasite intensity (n, number of dissected mosquitoes; statistical descriptions of infections are presented in Supplementary Tables S2 and S3). doi:10.1371/journal.ppat.1000576.g001

falciparum, the phenotype of *dsAPL1A* controlled infection prevalence, not parasite intensity. Knockdowns of *APL1B* or *APL1C* produced phenotypes indistinguishable from the *dsGFP*-injected controls, indicating that the functions of these two paralogs have no effect on either *P. falciparum* prevalence of infection or parasite intensity.

The above results suggest that the paralog *APL1A* is solely responsible for the host protective effect observed in the *dswAPL1* comprehensive knockdown of the *APL1* gene family. Because the two dsRNAs were tested in different experimental contexts (*dswAPL1* against wild parasites in Cameroon, *dsAPL1A* against gametocyte culture in the lab), we directly compared them by dsRNA treatment followed by challenge with cultured gameto-

cytes in the same experimental infections. We verified that dswAPL1 silences all three APL1 paralogs, and that expression of APL1A was diminished with equivalent efficiency by treatment with either dswAPL1 or dsAPL1A (Figure 3A). After infection, both dswAPL1 and dsAPL1A produced equivalent, significantly higher *P. falciparum* infection prevalence than the dsGFP controls (Figure 3B; dsAPL1A, p = 0.0004; dswAPL1, p = 0.0007). This result confirms that of the APL1 family, only APL1A acts as a host protective factor to limit infection with *P. falciparum* derived either from the field population or from *in vitro* gametocyte culture.

Interestingly, APL1A protection appears to operate with comparable efficiency against a multiplicity of parasite genotypes. The replicates of the *dswAPL1* treatment were each infected with blood from a different naturally-infected gametocyte carrier in Cameroon, who harbored different parasite genotypes (data not shown), and yet the combined replicates of the *dswAPL1* treatment displayed significantly elevated infection prevalence. Individually, each of the replicates also displayed the same trend of increased prevalence, although not significantly so, probably due to the reduced sample size. In addition, the *dsAPL1A*-treated mosquitoes were challenged with cultured gametocytes of the NF54 strain, which also contains multiple genotypes [13,14]. Thus, the APL1A-mediated mechanism appears to protect mosquitoes against *P. falciparum* in a way that is probably general and not highly sensitive to parasite strain and genotype variation.

APL1C protects against P. berghei and P. yoelii infection

We previously demonstrated that APL1 family protection against the rodent malaria parasite P. berghei was mediated only by paralog APL1C in the G3 strain of A. gambiae [9]. In order to eliminate the formal possibility that the result obtained in G3 was a mosquito strain effect, we tested protection by the APL1 paralogs against P. berghei in the Ngousso mosquito strain. In contrast to the results obtained above for P. falciparum, and identically to the results obtained in the G3 strain for P. berghei, the APL1C paralog also strongly limits infection with P. berghei in Ngousso mosquitoes (Figure 4; infection prevalence, $p = 2.825 \times 10^{-7}$; oocyst intensity, p = 0.0004). In addition, we obtained an equivalent result showing APL1C protection against P. berghei in a third strain of A. gambiae, the Yaoundé strain (data not shown). Thus, APL1C-based protection against *P. berghei* appears to be a general phenomenon, and is not restricted to a particular mosquito strain-parasite genetic combination as would be expected if APL1C protection displayed a large genotype-by-genotype effect.

We examined APL1 protection in Ngousso mosquitoes against another species of rodent malaria parasite, P. yoelii, which typically infects A. gambiae with a mixture of normal and melanized oocysts [15]. The dsGFP-treated controls permitted a median of 1 normal P. yoelii oocyst with 46% infection prevalence for normal oocysts and 48% harboring melanized parasites, while dsAPL1C-treated mosquitoes permitted a median of 28 normal oocysts with 86% infection prevalence for normal oocysts and 1.5% harboring melanized parasites, (dsAPL1C infection prevalence for normal oocysts, $p = 6.2 \times 10^{-12}$; intensity of normal oocysts, $p = 1.7 \times 10^{-8}$; presence of melanized parasites, $p = 6.8 \times 10^{-7}$). The APL1A or APL1B knockdowns were not significantly different from dsGFP controls for normal oocyst infection prevalence or intensity. Thus, APL1C is required for control of infection prevalence and oocyst intensity against P. yoelii similar to P. berghei, but with the difference that at least part of the APL1C-dependent protection against P. yoelii is mediated by the melanization response, which in this case is also APL1C-dependent. Overall, these results indicate that among the APL1 paralogs, APL1A is exclusively responsible for establishing the permissive or non-permissive state for P. falciparum



Figure 2. Paralog APL1A limits P. falciparum development in A. gambiae. A) Injection of dsRNAs that individually target APL1A, APL1B, or APL1C specifically diminish transcript levels from the cognate genes. B) Silencing of APL1A but not APL1B or APL1C permits significantly higher infection prevalence in mosquitoes fed on *in vitro* cultured P. falciparum gametocytes, with no detectable effect on parasite intensity. doi:10.1371/journal.ppat.1000576.g002

infection, with no detectable effect on *P. berghei* or *P. yoelii* infection; while *APL1C* among the paralogs limits infection by *P. berghei* and *P. yoelii*, but has no detectable influence upon *P. falciparum* development.

Rel2 controls P. falciparum infection prevalence

In *Drosophila*, the Toll and Imd immune pathways are responsible for protection against a broad range of pathogens, including fungi, Gram-positive and Gram-negative bacteria [16]. When these pathways are activated, immune signals traverse multiple intracellular intermediates and are ultimately converted into regulated gene expression by the action of transcription factors Dif and Relish, respectively. Orthologs and homologs of some of these *Drosophila* factors have been identified in *A. gambiae* [17,18]. At the terminus of the *A. gambiae* Toll pathway, Rel1 (also known as Gambif1) serves as the functional analogue of *Drosophila* Dif (no direct Dif ortholog has been identified in *A. gambiae*), while Rel2 is orthologous to the *Drosophila* transcription factor Relish [19–21].

Protection of *A. gambiae* against infection by *P. berghei* is largely mediated by Rel1 [9,20], which translocates to the nucleus and activates transcription after the degradation of its cytoplasmic inhibitor, Cactus [22]. We previously showed that *APL1C* expression is required for Rel1-mediated protection against *P.*

berghei [9]. Here, we first asked whether Rel1 or Rel2-mediated immune signalling controls the development of P. falciparum in A. gambiae. We silenced the ultimate transcription factor of each pathway in the Ngousso mosquito strain, followed by challenge with cultured P. falciparum gametocytes. Only knockdown of Rel2, and not Rell, produced a phenotype, leading to a significant increase in infection prevalence compared to dsGFP-injected controls (Figure 5A&B, Rel2 p = 0.0002, Rel1 p = 0.908). Next, we constitutively activated Rel1 by silencing the gene for its inhibitor Cactus, followed by P. falciparum infection. The dsCactustreated mosquitoes, with constitutively activated Rel1, displayed no P. falciparum infection difference as compared to controls (Supplementary Figure S1). Together, these results indicate that the Rel2 pathway, and not Rel1, is responsible for anti-P. falciparum protection, which is the inverse of the immune signalling profile that protects mosquitoes against P. berghei infection [9,20].

The Rel2-Short isoform controls *APL1A* gene expression and *P. falciparum* infection

The infection prevalence switch from non-permissive to permissive state for *P. falciparum* in the Rel2 knockdown mosquitoes might result from the removal of an unknown intrinsic molecular barrier to *P. falciparum* development that is frequent in wild-type *A. gambiae*. Because the knockdowns of *Rel2* and *APL1A*



Figure 3. The effect of paralog *APL1A* **alone is sufficient to explain APL1 family protective function against** *P. falciparum.* **A)** RT-PCR of individual *APL1* paralogs confirms that *dswAPL1* efficiently silences all three *APL1* family members. Right panel verifies directly that *APL1A* is silenced with equivalent efficiency by either *dsAPL1A* or *dswAPL1*. **B)** Treatment of mosquitoes with either *dswAPL1* or *dsAPL1A* causes equivalent reduction in host protection against *P. falciparum* infection. doi:10.1371/journal.ppat.1000576.g003

displayed the same phenotype, that is, elevated infection prevalence, we hypothesized that APL1A might be a mediator of the *P. falciparum* developmental blockade imposed by Rel2. Comparison of APL1A and Rel2 effects shows that mosquitoes silenced for each gene displayed comparable *P. falciparum* infection prevalences of 78% (Figure 2B) and 81% (Figure 5B) respectively, where the controls in both experiments were also comparable (57% and 59% prevalence, respectively). Thus, by magnitude of effect, the function of APL1A appears sufficient in itself to explain the protection by Rel2 against *P. falciparum*, suggesting that APL1A could be an important mediator of Rel2 protection against this parasite species.

To further explore the relationship between APL1A and Rel2 in immune signalling, we first examined transcriptional regulation of APL1A. There are two isoforms of Rel2 in A. gambiae: a Full-length form (Rel2-F) and a Short form (Rel2-S) lacking the inhibitory ankyrin repeats and death domain present in the full-length protein [21]. The dsRel2 construct used above, which caused increased P. falciparum infection prevalence in treated mosquitoes (Figure 5B), targets a common region shared by both Rel2 isoforms. Gene silencing using the shared Rel2 construct also abolished transcript of APL1A, but not APL1B or APL1C (Figure 5C), indicating that transcript levels of paralog APL1A are under Rel2 control. To distinguish effects of the Rel2 isoforms we then used dsRel2(Ank), a knockdown construct that targets the ankyrin domain present only in the full-length Rel2-F transcript, which consequently silences only Rel2-F but not the shorter Rel2-S [21]. Because Rel2-F includes the sequences present in Rel2-S, it is not possible to design a Rel2-S specific knockdown. In contrast to the absence of APL1A mRNA after treatment with dsRel2, mosquitoes treated with dsRel2(Ank) displayed similar APL1A transcript levels as dsGFP controls (Figure 5C). Corresponding to the ability to control APL1A expression, dsRel2 treated mosquitoes displayed significantly increased P. falciparum infection prevalence while dsRel2(Ank) treated mosquitoes were not different from controls (Figure 5D). These results indicate that the Rel2-S and not Rel2-F isoform controls P. falciparum development in A. gambiae, and that the ability of a Rel2 isoform to control infection or not is correlated with its ability to regulate APL1A mRNA levels. There was no detectable effect of dsAPL1A, dsAPL1B, or dsAPL1C on Rel2 transcript levels (data not shown), suggesting that APL1A is downstream of Rel2-S in an immune regulatory network that has unidirectional polarity. Taken together, these results suggest that some or most of the anti-P. falciparum activity of Rel2, and specifically of the Rel2-S isoform, is dependent upon APL1A function, and that the mechanism of anti-parasite protection probably operates by Rel2-S transcriptional control over APL1A expression.

Discussion

Here we show that paralogs of a mosquito LRR gene family, *APL1*, display distinct activities against different species of the genus *Plasmodium*. Each *APL1* paralog is required for the full protection of mosquitoes against only one of the two malaria parasite lineages tested, and each *APL1* paralog protects by involvement in only one of two separate immune signalling pathways. Thus, paralog APL1A is required for protection against



Figure 4. Paralog *APL1C* **limits development of rodent malaria species in** *A. gambiae.* **A)** Silencing of *APL1C* but not *APL1A* or *APL1B* permits significantly higher oocyst intensity in mosquitoes fed on mice infected with *P. berghei.* **B)** Silencing of *APL1C* also permits significantly higher *P. berghei* infection prevalence. Increases in both parasite intensity and infection prevalence were observed for another rodent malaria parasite, *P. yoelii*, following silencing of *APL1C* but not *APL1A* or *APL1B* (see Results). doi:10.1371/journal.ppat.1000576.g004

human *P. falciparum*, but not rodent *P. berghei* or *P. yoelii*, via the Imd/Rel2 pathway, while APL1C is required for protection against *P. berghei* and *P. yoelii* but not *P. falciparum*, and functions in the Toll/Rel1 pathway. These observations indicate that at least two reciprocal and mutually exclusive host-defense mechanisms, defined by the requirement for distinct *APL1* paralogs, protect mosquitoes from infection by different lineages of *Plasmodium* parasites.

It is recognized that invertebrate innate immunity distinguishes between broad groups of pathogens based on wide taxonomic and molecular differences, for example Gram-positive versus Gramnegative bacteria [16]. However, the current results demonstrate a previously unrecognized fine level of pathogen discrimination in the innate immune response of mosquitoes. We show that pathogens in the same genus, displaying high morphological and genetic relatedness, and with nearly identical mechanisms of cell invasion and development in the mosquito host, are nevertheless recognized and/or processed by mechanisms that require the function of closely related but distinct members of the APL1 family. The APL1 paralogs that are required for Rel1 or Rel2-dependent host protection, respectively, likely result from gene duplication and divergence [9]. The current study shows mutually exclusive protective function for two of the three known APL1 paralogs, against three different pathogens in two evolutionary groups.

Overall, these data indicate that APL1A is required for full host protection against *P. falciparum*. This protection is dependent on

Rel2-S but is independent of Rel2-F (Figure 6). Further work will be necessary to determine the structure of genetic and protein signaling in the Rel2-F subcircuit that regulates APL1-dependent host-protection, the mechanism of that protection, and the patterns of natural selection that shaped the APL1 family. Recent work reported the requirement of APL1C for stability of a protein complex that includes the complement-related protein TEP1 and the LRR protein LRIM1 [23,24]. The protein complex was tested only for protection against P. berghei and not P. falciparum, but in any case an APL1C-dependent complex could only be relevant for protection against the rodent parasite, because as shown here APL1C does not participate in protection against P. falciparum, and also because the other APL1C complex member, LRIM1, was previously shown to lack effect against P. falciparum [25]. Thus there is currently no reason to assume that APL1A-mediated protection against P. falciparum would be mechanistically similar to APL1C function in a soluble protein complex, particularly given their respective action in distinct immune signaling axes. We further found by functional analysis that APL2 (also known as LRRD7 [11]), another LRR-encoding candidate gene located within the PRI locus for which we found no protective effect against P. berghei [4], also had no significant effect upon P. falciparum development (data not shown).

These results do not imply that the functions of *APL1A* and *APL1C* are to protect exclusively against *P. falciparum* and the rodent malaria lineage, respectively, in a "gene-for-pathogen"



Figure 5. The major *A. gambiae* **defense against** *P. falciparum* **is controlled by the short isoform of the immune signalling factor Rel2, which regulates** *APL1A* **expression. A)** Gene silencing of immune signalling factors Rel1 and Rel2 is specific for the targeted genes. **B**) Silencing of *Rel2* but not *Rel1* permits significantly higher *P. falciparum* infection prevalence in treated mosquitoes. **C)** Genetic interactions between *Rel2* isoforms and *APL1* paralogs. *dsRel2* targets a common region shared by *Rel2-Short* and the ankyrin-domain containing *Rel2-Full* transcript isoforms. *dsRel2* treatment diminishes both of the *Rel2* transcripts, and also specifically silences *APL1A* but not *APL1B* or *APL1C* (RT-PCR in the horizontal row labelled Rel2 detects both *Rel2* transcripts, while the row labelled Rel2(Ank) detects only the longer *Rel2-Full* form). In contrast, treatment with *dsRel2(Ank)* efficiently diminishes the cognate long-form transcript (RT-PCR in row labelled Rel2(Ank)) and presumably the long-form proportion of the total *Rel2* transcripts detected by Rel2 RT-PCR (row Rel2), but does not affect transcript levels of *APL1A, APL1B, or APL1C*. **D**) Of the *Rel2* isoforms, only the short form *Rel2-S* is required both for *APL1A* expression and for protection against *P. falciparum* infection. *dsRel2* treatment causes significantly higher no occyst intensity. *dsRel2(Ank)* targets only the long-form transcript, and does not affect *APL1A* levels nor influence parasite infection.

system. Rather, it seems more likely that the function of the *APL1* paralogs illuminates different host-defense programs that act against repertoires of pathogens based on some shared characteristics, in other words, a "gene-for-pathogen-class" system. On this idea, APL1A protects not only against *P. falciparum* but also against as yet unknown "A-like" pathogens, and similarly for APL1C and "C-like" pathogens. We demonstrate here that activity of the correct APL1 factor is required to orient host defense towards human or rodent malaria parasites, respectively. There could be other gene families with similar fine-grained function against other pathogen classes, together yielding a potentially large combinatorial spectrum of protection profiles.

It is worth noting that APL1C protection against *P. berghei* and *P. yoelii* can not be the product of shared evolutionary history between the mosquito host and pathogen, because *A. gambiae* is not a natural vector of these parasites [26]. Thus, for *A. gambiae*, every experimental laboratory infection with rodent malaria represents exposure to a novel pathogen, and consequently the pathogen-resistance mechanisms in *A. gambiae* that are capable of protecting against rodent malaria species must have been shaped and maintained by interactions with other pathogens, and not by

rodent malaria. Consequently, it would be interesting to know which natural pathogens of *Anopheles* are in the "C-like" group. In addition, does an APL1C ortholog protect the natural vector of *P. berghei*, *A. dureni* [26], against infection, and does it also function by a Rel1-dependent mechanism? In contrast, *A. gambiae* does have a shared evolutionary history with *P. falciparum*, and thus in principle host resistance mechanisms, including *APL1A*, could have been selected by that pathogen, although to date this has not been proven.

Further work will be required to distinguish between a variety of potential explanations for the reciprocal and mutually exclusive nature of *APL1* paralog protection displayed against different pathogens. For example, APL1 proteins could be soluble factors that recognize parasite surface molecular patterns. Their high level of structural polymorphism [9] could be consistent with tracking pathogen variation, but at least APL1C does not appear to detectably bind to the *P. berghei* surface [24] (and data not shown). Alternatively, APL1 proteins could be components of immune signaling nodes, where their activity against a given pathogen class might reflect transduction of upstream class-specific immune signals, or differential sensitivity to attack by pathogen-derived



Figure 6. Model of APL1 paralog protection against pathogen classes in A. gambiae. In the cartoon, the rodent malaria parasites P. berghei and P. voelii, which are experimental but not natural pathogens of A. gambiae, exemplify a class of pathogens whose development in the host is inhibited by an APL1C-dependent mechanism. The "C-like" class of pathogens against which APL1C protects should include other unknown natural A. gambiae pathogens. The APL1C pathway is depicted superimposed on the Toll pathway because the Cactus/Rel1 module regulates levels of APL1C transcript and protein [9], but the presumptive extracellular signals that elicit APL1C protection are unknown, as are the immune receptors and the signalling intermediates except Cactus/Rel1. P. falciparum protection is mediated also by unknown immune receptors and signalling intermediates, and is shown superimposed on the Imd pathway because part of the Rel2 signaling module is involved. Specifically, the short Rel2 isoform, Rel2-S, which contains the Rel-homology domain (RHD) is required for host protection against P. falciparum infection and for APL1A gene expression. In contrast, the long Rel2 isoform, Rel2-F, does not influence protection against P. falciparum or APL1A transcript levels, and thus is depicted as a functionally unrelated shunt from the APL1A-terminated pathway

doi:10.1371/journal.ppat.1000576.g006

virulence factors. Here, the observed high rate of structural polymorphism might be driven by the need to escape inactivation by virulence factors, as hypothesized for the Relish cleavage complex in *Drosophila* [27,28]. Further dissecting the theme of potential APL1 involvement in signaling rather than direct pathogen detection, APL1 gene products might function as guard proteins, in the terminology of plant host-defense studies [29], where the guard protein initiates an immune alarm if its binding partner is subjected to virulence factor attack. In this case, the factors guarded by APL1 proteins against virulence factor attack might be Rel1 and/or Rel2, which could be consistent with the requirement of APL1 activity for respective Rel function. Finally, it is possible that APL1 proteins might serve only as sensors of pathogen virulence factors, akin to the still-hypothetical plant Decoy Model [30]. On this model, the APL1 proteins would be under the functional constraints to i) serve as a molecular mimic of the actual virulence factor target in order to provoke a virulence factor attack, and ii) interact with a downstream host immunesignaling factor in order to transduce an immune signal in the event that the virulence factor is detected. In this case, the system of APL1 paralogs and their structural variants might be explained as the product of balancing selection to remain attractive for attack by an evolving virulence factor.

There is not yet a large enough dataset of pathogen-host interactions in mosquito, especially with natural pathogens like P. falciparum, to be able to distinguish between these or other possible models of APL1 protection. Indeed, to our knowledge this report represents the first time that even three pathogens have been used in a detailed functional study of mosquito host-defense. One current need is for repeated observations using distinct pathogens and different host genetic backgrounds. The phenotypic complexity of such studies requires better measures of phenotypic difference. The standard approach has been pooling of results from three replicate gene knockdown experiments to statistically test the effect of a given dsRNA, but simple pooling is not statistically valid if there are significant differences between replicates of the same treatment. Instead, here we first perform statistical comparison of replicates to determine whether pooling is appropriate, and if it is not (which can occur for many reasons), we use meta-analysis to compare treatment effects. Such an approach, combined with the use of a stringent significance cut-off (here we use p = 0.01), should increase comparability between experiments and between results from different laboratories.

The APL1 genes were discovered as candidate factors within the PRI genetic locus that displays strong control over P. falciparum infection in nature [4]. In the current work we establish a validation trail that proceeds from that incriminated trait locus to the laboratory. We first functionally assayed APL1 candidate gene protection in Cameroon against wild P. falciparum, then verified the protective phenotype using the same mosquito strain and dsRNA but with laboratory-cultured P. falciparum gametocytes. We were then able to functionally dissect some of the genes and signalling pathways underlying the original Cameroon result but under controlled laboratory conditions. We emphasize that the work presented herein does not prove that APL1 underlies the phenotypic variation controlled by the PRI locus as detected in nature. For gene incrimination, it would be necessary at least to demonstrate a genetic association signal from the wild population. Thus, APL1A remains only a candidate gene, although strengthened as an interesting genetic candidate because of its functional properties as reported here. In contrast, based on their lack of functional effect, paralogs APL1B and APL1C, as well as the unrelated LRR protein APL2, can probably be excluded as genetic candidates for natural control of P. falciparum infection.

One interesting observation is that *APL1A* and *Rel2-S* knockdown phenotypes control *P. falciparum* infection prevalence at the usual infection levels that were obtained using either natural or *in vitro* parasites. This may be consistent with the previous speculation based on the genetic behaviour of the PRI locus in the wild population that mosquito resistance to *P. falciparum* could be the wild-type state, and that susceptibility might result from major loss-of-function mutations for an aspect of immune protection [4]. The binary infection prevalence switch observed here might have the features of such a gain- or loss-of-function system. It is also worth noting that, in terms of malaria transmission control, a mechanism that reduces infection prevalence should have more impact than reducing parasite numbers, because even one surviving oocyst is adequate for transmission.

Methods

Ethics statement

The use of human subjects was approved by the National Ethical Committee of Cameroon (protocol #039/CNE/MP/06). A signed individual written informed consent agreement was obtained from each subject or in the case of children from their parent or legally authorized representative before enrolment in the study.

Mosquitoes

Mosquitoes were reared according to standard procedures [9]. The Ngousso strain was originally initiated with mosquitoes collected in Yaoundé, Cameroon in January 2006, and belongs to the M molecular and Forest chromosomal forms. The strain was reared in insectaries of OCEAC, Cameroon, or the CEPIA mosquito production facility, Institut Pasteur, France.

Double-stranded RNA synthesis and injection

Double-stranded RNAs were synthesized from PCR amplicons using the T7 Megascript Kit (Ambion) as described previously [9]. The sequences of primers used for synthesis of dsRNA templates are in Supplementary Table S1. For each targeted gene, 200 ng of dsRNA (up to 207 nl volume, depending on the concentration) was injected into the thorax of cold-anesthetized 1–2 d-old *A.* gambiae females using a nano-injector (Nanoject II, Drummond Scientific). dsGFP or dsLacZ were used as controls. Injections were performed with a glass capillary needle as previously described [9,31]. Mosquitoes were challenged with malaria parasites by feeding on the appropriate bloodmeal 4 d after dsRNA injection.

Gene knockdown verification

The efficiency of the gene knockdown effect was monitored 4 d after dsRNA injection. cDNA synthesis was performed by using the M-MLV reverse transcriptase and random hexamers (Invitrogen). In each case, 500 ng of total RNA was used in triplicate assays. The triplicates were pooled and the mixture was used as template for PCR analysis. The sequences of primers used for gene knockdown verification are in Supplementary Table S1.

Experimental infection of mosquitoes with wild *P*. *falciparum*

Gametocyte carriers were identified upon reading of thick blood smears from 5 to 11 year old children from the Mfou district, 30 km from Yaoundé, Cameroon. Carriers harboring over 20 gametocytes/ μ l were enrolled upon signature of an informed consent form by their legal guardian. Blood was collected by venipuncture in heparinized Vacutainer tubes, centrifuged at 2000 g for 5 minutes at 37°C, and the supernatant containing patient serum was replaced by non-immune AB serum. A blood volume of 350 μ l was dispensed in glass feeders maintained at 37°C, and 5–6 d old female Ngousso strain mosquitoes were allowed to feed through a Parafilm membrane for 20 min. Only fully fed mosquitoes were retained for further analysis. Mosquitoes were maintained in an insectary at 27°C, 85% relative humidity on an 8% sucrose diet. The use of human subjects was approved by the National Ethical Committee of Cameroon.

P. falciparum gametocyte culture and mosquito infection

P. falciparum isolate NF54 [32,33] was cultured using the automated tipper-table system of Ponnudurai et al [34] as implemented in the CEPIA mosquito infection facility of Institut Pasteur. Briefly, a subculture of thawed NF54 stabilate was grown

in 10 ml RPMI 1640 medium supplemented with 25 mM HEPES and L-glutamine, 10% heat-inactivated human serum, and sodium bicarbonate at 0.2% final concentration under a constant gas regime (5% CO2, 1% O2, 94% N2). Fresh anonymous erythrocytes obtained from blood banks were added to 7% final concentration. Fourteen days after initiating the subculture, gametocyte maturity was tested by exflagellation of microgametes, and parasitemia and numbers of mature male and female gametocytes were counted on Giemsa stained slides.

Ten ml of culture was centrifuged at 2000 rpm, and the cell pellet was resuspended in an equal volume of normal type AB human serum. The infected erythrocytes were added to fresh erythrocytes in AB human serum and were transferred to a membrane feeder warmed to 37° C. Mosquitoes were allowed to feed for 15 minutes, and only fully engorged females were used for further analysis. Bloodfed mosquitoes were maintained on 10% sucrose solution supplemented with 0.05% para-amino benzoic acid.

Rodent malaria infection

Mosquitoes were fed on mice infected with *P. berghei* strain PbGFPCON [35], which constitutively expresses green fluorescent protein (GFP), or with *P. yoelii yoelii* strain 17XNL at 8–12% parasitemia with mature gametocytes. Mosquitoes were maintained at 21°C (*P. berghei*) or 24°C (*P. yoelii*) and 70% relative humidity on 10% sucrose supplemented with 0.05% para-amino benzoic acid.

Analysis of phenotypes

Midguts were dissected at 7–8 d post-infection, and for P. *falciparum* or P. *yoelii*: stained in 0.4% mercurochrome and the number of oocysts counted by light microscopy; or for P. *berghei*: no mercurochrome and counting by fluorescence microscopy. A mosquito was considered infected if it harbored at least one oocyst. Prevalence of infection is defined as the proportion of infected mosquitoes among the total dissected mosquitoes. Infection intensity is defined as the number of oocysts per mosquito, determined using mosquitoes harboring at least one oocyst. It is necessary to exclude the uninfected fraction of mosquitoes from analysis of intensity, because otherwise the result confounds infection prevalence and parasite intensity effects.

For each gene-silencing experiment at least 30 surviving mosquitoes were counted for oocyst load, and at least three independent replicate experiments were performed per experimental treatment. For statistical analysis, first all experiments were tested for statistical differences within treatments across replicates. If no differences between replicates were detected, replicates were pooled and differences in prevalence were tested using Chi Square and differences in intensity were tested using non-parametric Wilcoxon Mann-Whitney or Kruskal Wallis ANOVA on Ranks. If differences between replicates were detected, which can occur due to the inherent variability of Plasmodium experimental infections, replicates were analyzed independently using the tests described above and p values from independent tests of significance were combined using the meta-analytical approach of Fisher [36], and this combined p value is reported here. By either approach, the threshold for significance was defined as p = 0.01.

Supporting Information

Table S1 Primers used for synthesis of double-stranded RNAs (prefix T7, T7 portion underlined) and for knockdown verification of target genes (suffix V). Final suffix indicates forward, F, or reverse, R, sense of primers. Where no V primers are listed (Rell,

Rel2, Rel2(Ank), and Cactus), the synthesis primers without T7 sequences were also used for verification.

Found at: doi:10.1371/journal.ppat.1000576.s001 (0.04 MB DOC)

Table S2 Infection prevalence, measured as the fraction of mosquitoes with at least one midgut oocyst, analyzed using Chi Square. Oocyst intensity is analyzed only in mosquitoes with ≥1midgut oocyst and is analyzed using non-parametric Wilcoxon Mann Whitney (WMW) tests. Analyses are presented by the figure number in which the data appears. Values lower than the significance threshold of p = 0.01 are shown in bold. 1: p values from pooling of replicate experiments for statistical analysis where there were not significant differences between replicates within treatments. 2: p values from independent analyses of replicate experiments and combination of independent p values using the meta-analysis method of Fisher (see Methods). This approach was used when there were significant differences within treatment across replicates.

Found at: doi:10.1371/journal.ppat.1000576.s002 (0.05 MB DOC)

References

- 1. World Health Organization http://www.who.int/tdr/diseases/malaria.
- Sinkins SP, Gould F (2006) Gene drive systems for insect disease vectors. Nat Rev Genet 7: 427–435.
- Thomas MB, Read AF (2007) Can fungal biopesticides control malaria? Nat Rev Microbiol 5: 377–383.
- Riehle MM, Markianos K, Niare O, Xu J, Li J, et al. (2006) Natural malaria infection in Anopheles gambiae is regulated by a single genomic control region. Science 312: 577–579.
- Niare O, Markianos K, Volz J, Oduol F, Toure A, et al. (2002) Genetic loci affecting resistance to human malaria parasites in a west african mosquito vector population. Science 298: 213–216.
- Menge DM, Zhong D, Guda T, Gouagna L, Githure J, et al. (2006) Quantitative trait loci controlling refractoriness to Plasmodium falciparum in natural Anopheles gambiae mosquitoes from a malaria-endemic region in western Kenya. Genetics 173: 235–241.
- Riehle MM, Markianos K, Lambrechts L, Xia A, Sharakhov I, et al. (2007) A major genetic locus controlling natural Plasmodium falciparum infection is shared by East and West African Anopheles gambiae. Malar J 6: 87.
- Padmanabhan M, Cournoyer P, Dinesh-Kumar SP (2008) The leucine rich repeat domain in plant innate immunity: a wealth of possibilities. Cell Microbiol.
- Richle MM, Xu J, Lazzaro BP, Rottschaefer SM, Coulibaly B, et al. (2008) Anopheles gambiae APL1 is a family of variable LRR proteins required for Rel1-mediated protection from the malaria parasite, Plasmodium berghei. PLoS ONE 3: e3672. doi:10.1371/journal.pone.0003672.
- Mendes AM, Schlegelmilch T, Cohuet A, Awono-Ambene P, De Iorio M, et al. (2008) Conserved mosquito/parasite interactions affect development of Plasmodium falciparum in Africa. PLoS Pathog 4: e1000069. doi:10.1371/journal. ppat.1000069.
- Dong Y, Aguilar R, Xi Z, Warr E, Mongin E, et al. (2006) Anopheles gambiae immune responses to human and rodent Plasmodium parasite species. PLoS Pathog 2: e52. doi:10.1371/journal.ppat.0020052.
- Tahar R, Boudin C, Thiery I, Bourgouin C (2002) Immune response of Anopheles gambiae to the early sporogonic stages of the human malaria parasite Plasmodium falciparum. EMBO J 21: 6673–6680.
- Frank M, Kirkman L, Costantini D, Sanyal S, Lavazec C, et al. (2008) Frequent recombination events generate diversity within the multi-copy variant antigen gene families of Plasmodium falciparum. Int J Parasitol 38: 1099–1109.
- Gomez-Saladin E, Fryauff DJ, Taylor WR, Laksana BS, Susanti AI, et al. (1999) Plasmodium falciparum mdr1 mutations and in vivo chloroquine resistance in Indonesia. Am J Trop Med Hyg 61: 240–244.
- Vaughan JA, Hensley L, Beier JC (1994) Sporogonic development of Plasmodium yoelii in five anopheline species. J Parasitol 80: 674–681.
- Lemaitre B, Hoffmann J (2007) The Host Defense of Drosophila melanogaster. Annu Rev Immunol.
- Christophides GK, Zdobnov E, Barillas-Mury C, Birney E, Blandin S, et al. (2002) Immunity-related genes and gene families in Anopheles gambiae. Science 298: 159–165.
- Waterhouse RM, Kriventseva EV, Meister S, Xi Z, Alvarez KS, et al. (2007) Evolutionary dynamics of immune-related genes and pathways in disease-vector mosquitoes. Science 316: 1738–1743.

Table S3 Descriptive results of infections used in corresponding figures as indicated. Data are shown from pooled replicate infections. Median and range shown are for mosquitoes with at least one oocyst. Proportions of infected mosquitoes (infection prevalence) are shown in the figures.

Found at: doi:10.1371/journal.ppat.1000576.s003 (0.03 MB DOC)

Figure S1 Constitutive activation of Rel1 by silencing of Cactus does not influence *P. falciparum* development. A) *dsCactus* treatment efficiently silences cactus gene expression. B) The Cactus-depleted state, previously shown to constitutively activate Rel1 (see Results) does not affect the efficiency of *P. falciparum* development in *A. gambiae*.

Found at: doi:10.1371/journal.ppat.1000576.s004 (3.46 MB TIF)

Author Contributions

Conceived and designed the experiments: CM MMR KDV CB. Performed the experiments: CM JCJ IT IM SEN. Analyzed the data: CM MMR EB IM KDV. Contributed reagents/materials/analysis tools: CM JX IM CB. Wrote the paper: CM MMR JX EB KDV CB.

- Barillas-Mury C, Charlesworth A, Gross I, Richman A, Hoffmann JA, et al. (1996) Immune factor Gambifl, a new rel family member from the human malaria vector, Anopheles gambiae. EMBO J 15: 4691–4701.
- Frolet C, Thoma M, Blandin S, Hoffmann JA, Levashina EA (2006) Boosting NF-kappaB-Dependent Basal Immunity of Anopheles gambiae Aborts Development of Plasmodium berghei. Immunity 25: 677–685.
- Meister S, Kanzok SM, Zheng XL, Luna C, Li TR, et al. (2005) Immune signaling pathways regulating bacterial and malaria parasite infection of the mosquito Anopheles gambiae. Proc Natl Acad Sci U S A.
- Belvin MP, Jin Y, Anderson KV (1995) Cactus protein degradation mediates Drosophila dorsal-ventral signaling. Genes Dev 9: 783–793.
- Povelones M, Waterhouse RM, Kafatos FC, Christophides GK (2009) Leucinerich repeat protein complex activates mosquito complement in defense against Plasmodium parasites. Science 324: 258–261.
- Fraiture M, Baxter RH, Steinert S, Chelliah Y, Frolet C, et al. (2009) Two mosquito LRR proteins function as complement control factors in the TEP1mediated killing of Plasmodium. Cell Host Microbe 5: 273–284.
- Cohuet A, Osta MA, Morlais I, Awono-Ambene PH, Michel K, et al. (2006) Anopheles and Plasmodium: from laboratory models to natural systems in the field. EMBO Rep 7: 1285–1289.
- Van Den Berghe L (1954) The history of the discovery of Plasmodium berghei. Indian J Malariol 8: 241–243.
- Begun DJ, Whitley P (2000) Adaptive evolution of relish, a Drosophila NFkappaB/IkappaB protein. Genetics 154: 1231–1238.
- Pal S, St Leger RJ, Wu LP (2007) Fungal peptide Destruxin A plays a specific role in suppressing the innate immune response in Drosophila melanogaster. J Biol Chem 282: 8969–8977.
- 29. Jones JD, Dangl JL (2006) The plant immune system. Nature 444: 323-329.
- Shabab M, Shindo T, Gu C, Kaschani F, Pansuriya T, et al. (2008) Fungal effector protein AVR2 targets diversifying defense-related cys proteases of tomato. Plant Cell 20: 1169–1183.
- Blandin S, Moita LF, Kocher T, Wilm M, Kafatos FC, et al. (2002) Reverse genetics in the mosquito Anopheles gambiae: targeted disruption of the Defensin gene. EMBO Rep 3: 852–856.
- Delemarre-van de Waal HA, van de Waal FC (1981) Een tweede patient met malaria tropica op natuurlijke wijze verkregen in Nederland. Ned T Geneesk 125: 375–377.
- Chulay JD, Schneider I, Cosgriff TM, Hoffman SL, Ballou WR, et al. (1986) Malaria transmitted to humans by mosquitoes infected from cultured Plasmodium falciparum. Am J Trop Med Hyg 35: 66–68.
- Ponnudurai T, Lensen AH, Leeuwenberg AD, Meuwissen JH (1982) Cultivation of fertile Plasmodium falciparum gametocytes in semi-automated systems. 1. Static cultures. Trans R Soc Trop Med Hyg 76: 812–818.
- Franke-Fayard B, Trueman H, Ramesar J, Mendoza J, van der Keur M, et al. (2004) A Plasmodium berghei reference line that constitutively expresses GFP at a high level throughout the complete life cycle. Mol Biochem Parasitol 137: 23–33.
- 36. Fisher RA (1932) Statistical Methods for Research Workers. London: Oliver and Boyd.